Author's response to reviews

Title: Timeliness of contact tracing among flight passengers for influenza A/H1N1 2009

Authors:

Corien M. Swaan (corien.swaan@rivm.nl)
Rolf Appels (RAppels@ggdkennemerland.nl)
Mirjam E.E. Kretzschmar (mirjam.kretzschmar@rivm.nl)
Jim E. van Steenbergen (jim.van.steenbergen@rivm.nl)

Version: 7 Date: 19 December 2011

Author's response to reviews: see over
Dear editors,

Hereby I kindly provide you with our feedback on the second comment round of the associate editor of the article “Timeliness of contact tracing among flight passengers for influenza A/H1N1 2009”.

Q1: C. M. Swaan was captured as corresponding author. Please check if correct.
This is correct.

Q2: Journal standard instruction for the article structure requires the section 'Authors' contributions' and 'Competing interest' for article type 'Research'; however, none was provided in the manuscript. Please check.
Authors’ contributions, Competing interests and Acknowledgements were inserted in the manuscript.

Q3: The header 'Background' was inserted in the Abstract section as it is required per journal’s standard instruction. Please check if appropriate.
Appropriate.

Kind regards, also on behalf of my coauthors,

Corien Swaan
Consultant Communicable Disease Control
Preparedness & Response Unit (LCI)
Center for Infectious Disease Control (Cib)
National Institute of Public Health and the Environment (RIVM)
The Netherlands

Corresponding address: corien.swaan@rivm.nl